

W&M ScholarWorks

**VIMS Articles** 

Virginia Institute of Marine Science

10-16-2020

# How Old Are Marshes on the East Coast, USA? Complex Patterns in Wetland Age Within and Among Regions

Anna E. Braswell

James B. Heffernan

Matthew L. Kirwan Virginia Institute of Marine Science

Follow this and additional works at: https://scholarworks.wm.edu/vimsarticles

Part of the Geomorphology Commons

#### **Recommended Citation**

Braswell, Anna E.; Heffernan, James B.; and Kirwan, Matthew L., How Old Are Marshes on the East Coast, USA? Complex Patterns in Wetland Age Within and Among Regions (2020). *Geophysical Research Letters*, 47(19), e2020GL089415. DOI: 10.1029/2020GL089415

This Article is brought to you for free and open access by the Virginia Institute of Marine Science at W&M ScholarWorks. It has been accepted for inclusion in VIMS Articles by an authorized administrator of W&M ScholarWorks. For more information, please contact scholarworks@wm.edu.



## **Geophysical Research Letters**

#### **RESEARCH LETTER**

10.1029/2020GL089415

#### **Key Points:**

- Marshes largely formed after SLR slowed (after 4,000 ybp). A second formation period occurred (~400 ybp), coinciding with land cover change
- Studies that dated marshes to assess historic sea-level rise had an older and less varied age distribution than other studies
- Coastal marshes are younger nearer the coast and in regions where land use change may have delivered large amounts of sediment to the coast

#### **Supporting Information:**

- Supporting Information S1
- Table S1
- Data Set S1

#### Correspondence to:

A. E. Braswell, a.braswell@ufl.edu

#### Citation:

Braswell, A. E., Heffernan, J. B., & Kirwan, M. L. (2020). How old are marshes on the East Coast, USA? Complex patterns in wetland age within and among regions. *Geophysical Research Letters*, 47, e2020GL089415. https://doi.org/10.1029/2020GL089415

Received 25 JUN 2020 Accepted 11 SEP 2020 Accepted article online 22 SEP 2020

©2020. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

### How Old Are Marshes on the East Coast, USA? Complex Patterns in Wetland Age Within and Among Regions

Anna E. Braswell<sup>1,2</sup>, James B. Heffernan<sup>2</sup>, and Matthew L. Kirwan<sup>3</sup>

<sup>1</sup>Fisheries and Aquatic Sciences Program, School of Forest Resources and Conservation, University of Florida, Gainesville, FL, USA, <sup>2</sup>Nicholas School of the Environment, Duke University, Durham, NC, USA, <sup>3</sup>Virginia Institute of Marine Sciences, Gloucester Point, VA, USA

**Abstract** Sea-level dynamics, sediment availability, and marine energy are critical drivers of coastal wetland formation and persistence, but their roles as continental-scale drivers remain unknown. We evaluated the timing and spatial variability of wetland formation from new and existing cores collected along the Atlantic and Gulf coasts of the United States. Most basal peat ages occurred after sea-level rise slowed (after ~4,000 years before present), but predominance of sea-level rise studies may skew age estimates toward older sites. Near-coastal sites tended to be younger, indicating creation of wetlands through basin infilling and overwash events. Age distributions differed among regions, with younger wetlands in the northeast and southeast corresponding to European colonization and deforestation. Across all cores, wetland age correlated strongly with basal peat depth. Marsh age elucidates the complex interactions between sea-level rise, sediment supply, and geomorphic setting in determining timing and location of marsh formation and future wetland persistence.

**Plain Language Summary** Marshes, a type of coastal wetland, form under stable, slow rates of sea-level rise (SLR). Although many marshes along the East Coast of the United States formed after the most recent slowing of SLR (approximately 4,000–6,000 years ago), there is evidence that sediment erosion from colonial deforestation expanded marshes. This study uses sediment cores to determine when and where marshes formed to understand what drives modern marsh formation. We found that most marshes did form following SLR slowing after around 4,000 years ago. We also found evidence of marshes forming during European colonization in the northeastern and southeastern United States. Finally, we found recent formation of marshes closest to the coast, possibly driven by storm events and associated overwash. Through this study, we outline a novel way to easily determine marsh age and to better understand why marshes form through future research.

#### 1. Introduction

Tidal marsh systems are geologically young and ephemeral, reliant on a source of sediment, relatively calm tectonics, and stable rates of sea-level rise (SLR) (Day et al., 2011; Dokka et al., 2006; Kirwan et al., 2010; Reed, 1995). At the end of the last ice age, the Laurentide Ice Sheet melted, and Greenland Ice Sheet reduced in size, increasing sea level rapidly (Carlson et al., 2008; Tarasov & Peltier, 2002). Current tidal marshes and complexes are thought to have formed as SLR slowed after the last ice age and about 4,000–6,000 years before present (ybp; Engelhart & Horton, 2012; Redfield, 1965), but to our knowledge, no systematic assessment of marsh age has analyzed this history.

Sediment builds and maintains tidal marsh platforms (Day et al., 2011), and its availability is controlled by both upland supply and transport (and retention) along drainage networks. As SLR decelerated and wetlands established along the coast (Saintilan et al., 2020), sediment loads would have varied with watershed characteristics including catchment area, slope, and drainage network structure, as well as bedrock geology and glacial and climatic history (Bell & Laine, 1985; Gordon, 1979; Johnson & Fecko, 2008; Leopold et al., 1964; Meade, 1982; L. F. Phillips & Schumm, 1987; Sella et al., 2007). Sediment supply also reflects the balance of sediments and water from marine and inland sources (Braswell & Heffernan, 2019; Ganju et al., 2013; Yousefi Lalimi et al., 2020). Near-coastal storms can also impact landform stability and wetland dynamics (Cahoon, 2006; Davis, 1994; Leonardi et al., 2018). These terrestrial and marine drivers may shape regional-, estuary-, and local-scale formation and persistence of coastal wetlands. Humans can promote or suppress wetland formation by mobilizing sediment and altering drainage networks (Syvitski et al., 2005; Walling, 2006; Weston, 2014). Deforestation, agricultural development, and erosion accompanied European colonization of eastern North America seaboard of the United States through the 17th–19th centuries (Cronon, 2003; Meade, 1982; Pfaff, 2000). The timing of this agricultural intensification is important for the fate of mobilized sediment. The resulting downstream pulse of sediment may have remained stored in fluvial networks (Trimble, 1999; Walter & Merritts, 2008) or was delivered to coasts where it increased marsh building (Gell et al., 2009; Kirwan et al., 2011; Tweel & Turner, 2012). Some studies have estimated that coastal sediment delivery could have dramatically increased marsh extents, as evidenced by significant expansion of marshes along the Mississippi River birdfoot delta (Tweel & Turner, 2012) and at Plum Island, MA (Kirwan et al., 2011). The degree of regional variation in this anthropogenic expansion and its potential drivers is uncertain.

To determine how climatic, geomorphic and anthropogenic drivers impact wetland formation, we analyzed an extensive data set of new and published coastal tidal marsh basal ages, from coastal areas spanning local and regional gradients of upland and marine influence. We classified each core by physical location, region, and study objectives to address the questions: (1) When did tidal marshes form? (2) Where is historical deforestation and colonization correlated with wetland formation? This study expands the spatial scale of previous studies, extending regional and site-specific wetland formation projects over a continental coastline.

#### 2. Materials and Methods

#### 2.1. Core Database

To understand the spatial patterns of wetland formation, we collected literature that reported dating evidence (mainly radiocarbon) from basal layers of tidal wetlands or salt marshes. We searched for literature in the Web of Science (Web of Science Core Collection) using combinations of the keywords "marsh," "age," "dating," and "radiocarbon." Studies along the Atlantic and Gulf coasts that clearly dated core samples at or near the basal peat layer were included in this study, regardless of study aims. With each date in the study, we recorded study objectives (sea level or other), depth of sample, region, and various other descriptors of each core (Texts S1 and Data Set S1).

#### 2.2. Field Study: Site Description and Core Collection

To characterize local spatial variability in wetland age and more directly evaluate the role of riverine and marine influence, we collected 97 cores across five sites (Text S1; Table S1): two pairs of riverine- and marine-dominated wetlands (Ogeechee River, GA and Cape Romain, SC; Cape Fear River, NC and Masonboro Island, NC) and a fifth site (Barnstable Harbor, MA) that we paired with Plum Island, MA, where a previous study had intensively sampled basal peat age and core depth (Kirwan et al., 2011). As in that study, we dated basal peat in a subset of cores and determined the period of formation on the basis of regional depth-age relationships.

To distribute samples within each site, we researched the history of wetland areas in each site, delineated marsh polygons from the National Wetlands Inventory (U.S. Fish and Wildlife, 2014), and assigned random points within each area (ArcGIS 10.5). We used a Russian peat corer with extension rods (Aquatic Research Instruments) to collect 5-cm-diameter, 50-cm-long half-moon core sections. Sediment cores were collected to 10 cm past the basal peat layer or to point of refusal. We visually inspected and described any major stratigraphy during coring and photographed each section in the field. Cores were stowed in cold coolers and transported back to the laboratory (Grainger River Center, Duke University, NC), where they were stored at 4°C until analysis.

#### 2.3. Field Study: Laboratory Analysis

We analyzed bulk density and percent organic matter (Text S1). Along each core, we determined where the basal peat layer or formation of the marsh occurred based on these factors: where the organic matter rose above 10% by mass (moving up from the bottom of the core), refusal during coring, or core description (basal peat layer described in field) (Belknap & Kraft, 1977). Four cores spanning the range of depths at each site (eight among two regions of the Cape Fear site) were chosen for radiocarbon dating (24 cores total), based on the clarity of the basal peat layer from organic matter content or field description and distance from other radiocarbon dated cores. Samples were prepared (Text S1) and shipped to the National Ocean Sciences





**Figure 1.** Our core database covered literature with radiocarbon dates along the Atlantic and Gulf Coasts of the United States. These regions had heterogeneous drivers of wetland formation resulting in varying marsh ages. The map displays proxies for the various drivers used in the remainder of the paper (river:tide ratio, distance to the coast, region), as well as the dependent variable, wetland age in years before present.

Accelerator Mass Spectrometry Facility at Woods Hole Oceanographic Institution for standard acid-baseacid pretreatment and radiocarbon analysis. We determined the calendar age of each sample using OxCal software (OxCal v4.3) calibrated with chronological models (Table S2). We chose the IntCal13 curve to determine the calibrated radiocarbon date within confidence intervals (Reimer et al., 2013).

#### 2.4. Age-Depth Relationships

We used age-depth relationships to estimate the date of formation for undated cores at each of our six intensive sites. We pooled our own measurements with published cores to find regional age-depth relationships as characterized by linear relationships between log-transformed depth and basal peat age (Text S1). From these relationships, we estimated age and its 95% confidence interval for each undated core. We then assigned each core to an age category (pre- or post-European settlement) based on whether estimated ages were greater or less than 400 ybp; the few cores with confidence intervals that overlapped 400 ybp were classified as uncertain. This binning allowed us to assess regional patterns in wetland age while minimizing the influence of uncertainty around age estimates (Text S1).

#### 2.5. Analysis

We used multiple measures of marine and upland influence to determine how geomorphic setting affects the initiation and persistence of coastal wetlands. We used distance from the ocean as both a proxy of both potential frequency of storm overwash events and as a potential mediator of upland versus marine sediment supply (Text S1). Although we recognize that geomorphic setting is more complex across the coastline than these proxies illustrate, our data synthesis approach does not allow for the detailed differentiation between geomorphic settings due to time constraints and scientific bias. To determine the comparative influence of watershed and estuary on each wetland, we used river:tide ratio (RTR; Braswell & Heffernan, 2019), which compares the annual volumes of river runoff and tidal inflows into an estuary (Text S1). We used Mclust package (version 5.2.3; R 3.4) to assess modality of the age distributions across several variables, specifically to determine if there was a younger mode of marsh formation by region, distance from coast, or RTR.



#### 3. Results

Most of the 387 cores (from 43 papers) in our literature database (see Table S4) had marsh initiation dates in the millennia following the slowing of SLR (literature, mean:  $2,450 \pm 90$  SE (standard error) ybp; median: 2,166 ybp), but varying geomorphic drivers and biases from the literature affect patterns in age distribution. Among all dated cores (field plus literature), the age ranged from 0 to 10,819 ybp (Figure 1; database, *n*: 411, mean:  $2,767 \pm 98$  SE ybp; median: 2,591 ybp). SLR studies dominated the literature of tidal marsh ages (Figure 2a). Of the 387 database cores, 314 (approximately 81%) are associated with studies that had objectives related to geologic records of past sea level. These cores are predominately older, after the deceleration of SLR in the Holocene (Figure 2a; mean:  $3,111 \pm 110$  SE ybp; median: 3,101 ybp). The remainder (73 cores; approximately 19%) come from studies that addressed marsh natural history, rather than sea level. These cores spanned a wider distribution and had no clear mode (Figure 2a; mean:  $1,955 \pm 199$  SE ybp; median: 1,721 ybp).

Wetlands closer to the coast had more young cores and a wider range of wetland ages (Figures 2h–2l, 3, and S1) than cores further inland. Cores that were further from the coast (moving up river or up sound) had a narrow distribution of ages (Figures 2h and 3; Greater than 20 km from ocean; *n*: 116, mean:  $3,778 \pm 134$  SE ybp, median: 3,776 ybp), with older dates than cores that occur close to the coast (Figures 2k and 3; between 0.5 and 1 km from the ocean; *n*: 60, mean:  $1,901 \pm 228$  SE ybp, median: 1,401 ybp).

Our proxy for inland sediment, RTR, also elucidated timing of wetland formation. Cores with a larger RTR (more riverine water volume than estuarine volume) have a narrower distribution and older mean and median age than cores that have lower RTR (Figures 2b and S1; Log (RTR) = 0–2; *n*: 34, mean: 4,819  $\pm$  233 SE ybp, median: 4,736 ybp). As river to tide ratio decreases (toward more marine influence), the distribution widens and contains younger dates of formation (Figure 2b; Log (RTR) = -4 to -2; *n*: 109, mean: 2,292  $\pm$  193 SE ybp, median: 1,662 ybp). Our data points to sites with less relative river influence have younger marshes than sites with more river influence.

The distribution of core age differs among regions (Figures 1, 2c–2g, and S1), but all regions had the highest abundance of cores between 2,500 and 3,000 ybp (all dated cores; *n*: 411; mean: 2,767  $\pm$  98 SE ybp; median: 2,591 ybp; Figures 2c–2g). Based on BIC and ICL scores, the distribution of ages across all regions is best described by two clusters or modes of unequal variance (Mclust version 5.2.3; R 3.4; using all dates greater than 0). We find an older mode of the distribution (*n*: 274, mean: 3,829  $\pm$  94 SE ybp; median: 3,660 ybp) and a smaller mode of younger cores (*n*: 128, mean: 701  $\pm$  43 SE ybp; median: 615.5 ybp). We found that the northeast Atlantic Coast has a younger and wider distribution of wetland ages, best described by one, normal mode (*n*: 84, mean: 2,065  $\pm$  239 SE ybp; median: 1,446 ybp). The southeastern Atlantic Coast also reflected this trend, albeit with a better fit of seven modes of equal variance (oldest two modes: *n*: 33, mean: 4079  $\pm$  175 SE ybp; median: 4,114 ybp; youngest two modes: *n*: 20, mean: 133  $\pm$  17 SE ybp; median: 145 ybp). Other regions were also best described by two or more modes, but either had extremely small young modes (*n* = <5) or the youngest mode had ages > 400 ybp.

Age-depth relationships by region were sufficiently strong to differentiate wetland initiation as before or after European colonization (Figure 4; Table S3). Explanatory power of depth as a predictor of age varied from 0.61 to 0.7 across regions, with slopes (after log-log transformation) ranging from 0.83 to 1.9 (Table S3). We repeated the above analyses using the all the cores, including our field cores dated by regional age-depth relationships. These results reflect similar trends and drivers, particularly a secondary younger peak of wetland formation (Text S2; Figure S1). We do find that the age-depth estimated cores added younger cores and more information to data poor areas such as the southeast (Figure S1). In addition, these cores add resolution to the dating in our field sites, allowing for the determination of age patterns in heterogeneous areas (Figure 4). Although unconventional, these cores complement the results found by the database alone.

#### 4. Discussion

Our results provide the first continental coastline scale synthesis of evidence that the main, first-order driver of coastal marsh formation is the rate of SLR. After the disintegration of the Laurentide Ice Sheet (Carlson et al., 2008) and the stabilization of the Greenland ice sheet, melt water to the ocean slackened, decreasing the rate of SLR (Tarasov & Peltier, 2002). Although the timing of these events is not well constrained





**Figure 2.** (a) Sea-level rise studies dominated our core database, with older dates of formation than studies with other objectives. These studies (blue histogram) skewed our database toward older cores with longer records, as seen in this distribution of wetland ages by study objectives. (b) Timing of wetland formation varied with river:tide ratio (RTR), as a proxy for the relative influence of the watershed to the estuary. Wetlands with higher RTR typically formed earlier and have less variability in age than wetlands with low RTR. (c-g) Regionally, wetlands formed at different times, with the northeast and southeast having younger wetlands than other regions (c, Maine; d, northeast; e, Mid-Atlantic; f, southeast; g, Gulf Coast). (h-l) Wetland age varied with distance to the ocean (decreasing downward through panels), as a proxy for marine and storm impacts and basin infilling, with wetlands closer to the coast having younger (h, >20 km; i, 10–20 km; j, 5–10 km; k, 1 km to 500 m; l, <500 m). On all panels, vertical black line represents approximate time of European colonization.

(Engelhart & Horton, 2012; Gehrels, 2010), it is generally agreed that the rate of SLR slowed between 4,000 and 6,000 ybp (Engelhart & Horton, 2012). The decrease in the rate of SLR likely varied within this window across regions along the coast based on local conditions. Our results confirm that most wetlands in our core database and field study formed after this window (mean: 2,450 ybp). This finding supports previous work





**Figure 3.** Regional trends in age reflect differences in marine influence and basin infilling. Here, we consider core age by region (*x*-axis) and distance to the coast (darker red is farthest from the coast). Cores were younger within 500 m from the ocean. We found cores were older with distance from the ocean (particularly in the Mid-Atlantic and southeast), pointing to sediment delivery from rivers as an important driver of wetland formation.

showing that tidal marshes are unable to form and persist under periods of great sea-level change (such as transitions in glacial-interglacial cycles), which limits adequate sedimentation and macrophyte survival (Kirwan et al., 2010; Marani et al., 2010; Morris et al., 2002; Saintilan et al., 2020). Although we do find a peak of wetland formation, there is much variation in the date of formation regionally and across all cores. This finding is consistent with both regional variation in the rate SLR and differences in timescales of wetland formation due to basin and watershed morphology.

Age distributions of marshes varied as a function of upland or riverine influence, as measured by RTR (ratio of annual river discharge to tidal estuary volume). Although there are many drivers of wetlands (e.g., vegetation feedbacks; Marani et al., 2010; Morris et al., 2002), sediment supply is a major control of marsh platforms (Bouma et al., 2016; D'Alpaos et al., 2012; Yousefi Lalimi et al., 2020). Cores from sites with a higher RTR formed earlier and varied less in age than sites with a lower RTR (Figure 2b). Areas with higher RTR should have greater sediment delivery and build elevation faster, and therefore, those areas are more likely to have a date that corresponds with the deceleration of SLR. Areas with lower RTR may take longer to develop wetlands due to sediment limitation. Our results support this hypothesis and, furthermore, are consistent with the finding that high RTR areas also support greater abundance of wetlands (Braswell & Heffernan, 2019).

Our results support the hypothesis that "resetting events," such as intense storms, shape wetland age distributions, particularly along the back of barrier islands in lagoonal systems (500 m away from coast; Figures 2h–2l and 3). We found more, younger cores in sites closest to the coast and a wider distribution of age. Washover fans from storms can initiate wetland creation by increasing elevation in back barrier areas to a level that supports vegetation (Davis, 1994; Osgood & Zieman, 1993). These fans are variable in space and time, depending on storm intensity and geomorphology of the barrier, leading to variability in age along landforms close to the coast (Donnelly et al., 2004; Morton & Sallenger, 2003).

Basin infilling, where the areas of the estuary closest to the mouth of the river fill with sediment first (Roy, 1994), can also create a gradient of wetland ages from the river to the coastline (Gell et al., 2009;





**Figure 4.** Our field data show variability in depth by region, landform, and distance to the ocean. The field sites were arrayed by a gradient of upland influence and by region. Our Massachusetts site, (b) Barnstable Harbor was paired with (a) data from Kirwan et al. (2011) at Plum Island. The North Carolina sites compared a barrier island system, (c) Masonboro Island, to large river system, (d) Cape Fear River. We paired (e) Bulls Bay at Cape Romaine NWR with the mouth of the (f) Ogeechee River in Georgia. Regionally, North Carolina and Massachusetts displayed more variation than Georgia/South Carolina, which had less young (shallow) cores based on our regional fits and depth binning (see Texts S1).

Redfield, 1972). We found support for this hypothesis in that sites furthest from the coast, mostly within large estuaries such as Chesapeake Bay, had less variable formation dates and more cores that dated from around the time of the slowing of SLR (Figures 2h–2l and 3). We found additional substantiation for this result in the field data from places like Cape Fear River, NC and Plum Island, MA where variability in date of formation and younger ages of cores occurred close to the coast (Figures 2h–2l and 3).

Through the interaction of agricultural erosion, geomorphic setting, and infrastructure, we find that the regional history of agriculture could impact the formation of wetlands along the northeastern seaboard and barrier islands systems of the southeastern United States. We determined that the northeastern region of the Atlantic Coast (Figures 2d and 3) had some younger dates that correspond with timing of European agricultural expansion (Foster et al., 1998; Klein Goldewijk et al., 2010, 2011). Humans both increase and reduce the amount of sediment reaching the ocean through deforestation and erosion of agriculture and damming or impounding rivers (Syvitski et al., 2005). Erosion was particularly intense in the southeastern Piedmont (Trimble, 1974), yet the large coastal plain and Piedmont region likely trapped much of this sediment in floodplains, making transport to the coast slow or negligible at relevant time scales (J. D. Phillips, 1991, 1995; Phillips et al., 1993; Trimble, 1977; Walter & Merritts, 2008). The short or nonexistent coastal plain and smaller drainage networks could contribute in efficient transport of sediment to the coast in the northeast (Gordon, 1979; Kirwan et al., 2011). This region also experienced peak agricultural earlier, before the creation of major dams (Foster et al., 1998). Although some of the sediment was certainly trapped behind mill dams (Walter & Merritts, 2008), the geomorphic setting of this region may have contributed to sediment reaching the coast. Based on our results, we propose that pulses of anthropogenic sediment from erosion happened earlier and in more connected drainage networks than in other regions, creating younger wetlands and expanding marsh complexes (Figures 2d and 3). Although more evidence is needed to determine if this is a true signal of regional wetland expansion, or an artifact of studies in our database, it does compliment previous studies that have found evidence of wetland expansion after European colonization and land-use change (Hilgartner & Brush, 2006; Kirwan et al., 2011; Mattheus et al., 2009; Tweel & Turner, 2012). This finding points to a regional expansion of wetlands postcolonial period where sediment was easily conveyed to the ocean and not stored in the floodplain or behind a dam.

Understanding the influence of watershed and marine drivers on wetland formation is imperative for predicting their future persistence under lower sediment regimes, high rates of SLR, intense coastal storms, and anthropogenic stressors. Our findings on the timing and conditions of formation help elucidate two previously debated mechanisms of marsh persistence. First, recently formed marshes point to the importance of sediment in growing and maintaining marshes. With declining sediment loads (Falcini et al., 2012; Syvitski et al., 2005; Weston, 2014), marshes may contract back to a low sediment, precolonial deforestation state (Kirwan et al., 2011). Old, Holocene marshes, formed during periods of slow rates of SLR, highlight the importance of SLR in marsh formation and persistence (Crosby et al., 2016; Törnqvist et al., 2020) and complementing recent findings on the connection of mangrove formation to SLR (Saintilan et al., 2020). Our study integrates these perspectives, emphasizing that marsh age is tied to the interactions between SLR and sediment supply. Mismatches between the conditions of formation (higher sediment supply and lower rate of SLR) and the current environment are likely to influence how marshes respond to future stressors. Through understanding marsh age, we reveal the drivers of marsh persistence including geomorphic setting, terrestrial and marine sediment supply, and barrier island dynamics.

#### Data Availability Statement

Data sets for this research are included in this paper's supplementary information files and on Harvard Dataverse (Radiocarbon Dates: https://doi.org/10.7910/DVN/KDXIRN; Marsh Age Literature Database: https://doi.org/10.7910/DVN/GHK8AS). Please see supplemental reference file and excel file of data used in the core database.

#### References

Belknap, D. F., & Kraft, J. C. (1977). Holocene relative sea-level changes and coastal stratigraphic units on the northwest flank of the Baltimore Canyon Trough geosyncline. Journal of Sedimentary Research, 47(2), 610–629.

Bell, M., & Laine, E. P. (1985). Erosion of the Laurentide region of North America by glacial and glaciofluvial processes. Quaternary Research, 23(2), 154–174. https://doi.org/10.1016/0033-5894(85)90026-2

#### Acknowledgments

Financial support was provided by N.S. F. GLD grant (#1530233), GCA Coastal Wetland Scholarship, and a Duke Wetland Center Grant. Core collection permission was given by Georgia DNR, Ossabaw Island Foundation, USFWS Cape Romain NWR, SC DHEC, Fort Caswell, Fort Fisher State Recreation Area, North Carolina NERR, and the Town of Barnstable. Field and lab assistance was given by R. Needham, A. Moss, R. Cope, S. Ludwig-Monty, D. Meyer-Arrivillaga, D. Calderon-Arrieta, N. Sherman-Jollis, S. Mahanes, K. Robbins-Thompson, S. Cleaver, and B. Hassett. We thank A. B. Murray, M. Marani, D. Urban, and two anonymous reviewers for critique that improved this work.

Bouma, T. J., van Belzen, J., Balke, T., van Dalen, J., Klaassen, P., Hartog, A. M., et al. (2016). Short-term mudflat dynamics drive long-term cyclic salt marsh dynamics. *Limnology and Oceanography*, 61(6), 2261–2275. https://doi.org/10.1002/lno.10374

Braswell, A. E., & Heffernan, J. B. (2019). Coastal wetland distributions: Delineating domains of macroscale drivers and local feedbacks. *Ecosystems*, 22(6), 1256–1270. https://doi.org/10.1007/s10021-018-0332-3

Cahoon, D. R. (2006). A review of major storm impacts on coastal wetland elevations. *Estuaries and Coasts*, 29(6 a), 889–898. https://doi. org/10.1007/BF02798648

Carlson, A. E., LeGrande, A. N., Oppo, D. W., Came, R. E., Schmidt, G. A., Anslow, F. S., et al. (2008). Rapid early. Holocene deglaciation of the Laurentide ice sheet, 1(September), 620–624. https://doi.org/10.1038/ngeo285

Cronon, W. (2003). Changes in the Land: Indians, colonists and the ecology of New England. New York: Hill and Wang.

Crosby, S. C., Angermeyer, A., Adler, J. M., Bertness, M. D., Deegan, L. A., Sibinga, N., & Leslie, H. M. (2016). Spartina alterniflora biomass allocation and temperature: Implications for salt marsh persistence with sea-level rise. Estuaries and Coasts, 40(1), 213–223. https://doi. org/10.1007/s12237-016-0142-9

D'Alpaos, A., Da Lio, C., & Marani, M. (2012). Biogeomorphology of tidal landforms: Physical and biological processes shaping the tidal landscape. *Ecohydrology*, 5(5), 550–562. https://doi.org/10.1002/eco.279

Davis, R. A. (1994). Geology of Holocene barrier island systems. Berlin: Springer - Verlag. https://doi.org/10.1007/978-3-642-78360-9

Day, J., Ibáñez, C., Scarton, F., Pont, D., & Hensel, P. (2011). Sustainability of Mediterranean deltaic and lagoon wetlands with sea-level rise: The importance of river input. *Estuaries and Coasts*, *34*, 483–493. https://doi.org/10.1007/s12237-011-9390-x

Dokka, R. K., Sella, G. F., & Dixon, T. H. (2006). Tectonic control of subsidence and southward displacement of southeast Louisiana with respect to stable North America. *Geophysical Research Letters*, *33*, L23308. https://doi.org/10.1029/2006GL027250

Donnelly, J. P., Butler, J., Roll, S., Wengren, M., & Webb, T. (2004). A backbarrier overwash record of intense storms from Brigantine, New Jersey. Marine Geology, 210, 107–121. https://doi.org/10.1016/j.margeo.2004.05.005

Engelhart, S. E., & Horton, B. P. (2012). Holocene sea level database for the Atlantic Coast of the United States. *Quaternary Science Reviews*, 54, 12–25. https://doi.org/10.1016/j.quascirev.2011.09.013

Falcini, F., Khan, N. S., Macelloni, L., Horton, B. P., Lutken, C. B., McKee, K. L., et al. (2012). Linking the historic 2011 Mississippi River flood to coastal wetland sedimentation. *Nature Geoscience*, 5(11), 803–807. https://doi.org/10.1038/ngeo1615

Foster, D. R., Motzkin, G., & Slater, B. (1998). Land-use history as long-term broad-scale disturbance: Regional forest dynamics in central New England, (May 1997). *Ecosystems*, 1, 96–119.

Ganju, N. K., Nidzieko, N. J., & Kirwan, M. L. (2013). Inferring tidal wetland stability from channel sediment fluxes: Observations and a conceptual model. Journal of Geophysical Research: Earth Surface, 118, 2045–2058. https://doi.org/10.1002/jgrf.20143

Gehrels, R. (2010). Sea-level changes since the Last Glacial Maximum: An appraisal of the IPCC Fourth Assessment Report. Journal of Quaternary Science, 25(May 2009), 26–38. https://doi.org/10.1002/jqs

Gell, P., Fluin, J., Tibby, J., Hancock, G., Harrison, J., Zawadzki, A., et al. (2009). Anthropogenic acceleration of sediment accretion in lowland floodplain wetlands, Murray-Darling Basin, Australia. *Geomorphology*, 108(1-2), 122–126. https://doi.org/10.1016/j. geomorph.2007.12.020

Gordon, R. B. (1979). Denudation rate of central New England determined for estuarine sedimentation. *American Journal of Science*, 279(6), 632–642. https://doi.org/10.2475/ajs.279.6.632

Hilgartner, W. B., & Brush, G. S. (2006). Prehistoric habitat stability and post-settlement habitat change in a Chesapeake Bay freshwater tidal wetland, USA. *The Holocene*, *16*(4), 479–494. https://doi.org/10.1191/0959683606hl938rp

Johnson, P. A., & Fecko, B. J. (2008). Regional channel geometry equations: A statistical comparison for physiographic provinces in the eastern US. *River Research and Applications*, 24(6), 823–834. https://doi.org/10.1002/rra.1080

Kirwan, M. L., Guntenspergen, G. R., D'Alpaos, A., Morris, J. T., Mudd, S. M., & Temmerman, S. (2010). Limits on the adaptability of coastal marshes to rising sea level. *Geophysical Research Letters*, 37, L23401. https://doi.org/10.1029/2010GL045489

Kirwan, M. L., Murray, A. B., Donnelly, J. P., & Corbett, D. R. (2011). Rapid wetland expansion during European settlement and its implication for marsh survival under modern sediment delivery rates. *Geology*, 39(5), 507–510. https://doi.org/10.1130/ G31789.1

Klein Goldewijk, K., Beusen, A., & Janssen, P. (2010). Long-term dynamic modeling of global population and built-up area in a spatially explicit way: HYDE 3.1. *The Holocene*, 20(4), 565–573. https://doi.org/10.1177/0959683609356587

Klein Goldewijk, K., Beusen, A., Van Drecht, G., & De Vos, M. (2011). The HYDE 3.1 spatially explicit database of human-induced global land-use change over the past 12,000 years. *Global Ecology and Biogeography*, 20(1), 73–86. https://doi.org/10.1111/j.1466-8238. 2010.00587.x

Leonardi, N., Carnacina, I., Donatelli, C., Kamal, N., James, A., Schuerch, M., & Temmerman, S. (2018). Dynamic interactions between coastal storms and salt marshes: A review. *Geomorphology*, 301, 92–107. https://doi.org/10.1016/j.geomorph.2017.11.001

Leopold, L. B., Wolman, M. G., & Miller, J. P. (1964). Fluvial processes in geomorphology. San Francisco: W.H. Freeman and Company. Marani, M., D'Alpaos, A., Lanzoni, S., Carniello, L., & Rinaldo, A. (2010). The importance of being coupled: Stable states and catastrophic shifts in tidal biomorphodynamics. *Journal of Geophysical Research*, 115, F04004. https://doi.org/10.1029/2009JF001600

Mattheus, C., Rodriguez, A., & McKee, B. A. (2009). Direct connectivity between upstream and downstream promotes rapid response of lower coastal-plain rivers to land-use change. *Geophysical Research Letters*, 36, L20401. https://doi.org/10.1029/2009GL039995

Meade, R. H. (1982). Sources, sinks and storage of river sediment in the Atlantic drainage of the United States. *The Journal of Geology*, 90(3), 235–252. https://doi.org/10.1086/628677

Morris, J. T., Sundareshwar, P. V., & Nietch, C. T. (2002). Responses of coastal wetlands to rising sea level. *Ecology*, 83(10), 2869–2877. Retrieved from. https://doi.org/10.1890/0012-9658(2002)083%5B2869:ROCWTR%5D2.0.CO;2

Morton, R. A., & Sallenger, A. H. (2003). Morphological impacts of extreme storms on sandy beaches and barriers. *Journal of Coastal Research*, 19(3), 560–573.

Osgood, D. T., & Zieman, J. C. (1993). Spatial and temporal patterns of substrate physicochemical parameters in different-aged barrier island marshes. *Estuarine, Coastal and Shelf Science*, 37(4), 421–436. https://doi.org/10.1006/ecss.1993.1065

Pfaff, A. S. P. (2000). From deforestation to reforestation in New England, United States. In M. Palo & H. Vanhanen (Eds.), World forests from deforestation to Transition? (Two (pp. 67–82). Dordrecht: Kluwer Academic Publishers. https://doi.org/10.1007/978-94-010-0942-3 4

Phillips, J. D. (1991). Fluvial sediment budgets in the North Carolina Piedmont. Geomorphology, 4(3-4), 231-241. https://doi.org/10.1016/0169-555X(91)90006-V

Phillips, J. D. (1995). Decoupling of sediment sources in large river basins. IAHS Publications-Series of Proceedings and Reports-Intern Assoc Hydrological Sciences, (226), 11–16.



Phillips, J. D., Wyrick, M., Robbins, G., & Flynn, M. (1993). Accelerated erosion on the North Carolina coastal plain. Physical Geography, 14, 114–130. https://doi.org/10.1080/02723646.1993.10642471

Phillips, L. F., & Schumm, S. A. (1987). Effect of regional slope on drainage networks. Geology, 15(9), 813–816. https://doi.org/10.1130/ 0091-7613(1987)15<813:EORSOD>2.0.CO

Redfield, A. (1965). Ontogeny of a salt marsh estuary. Science, 147(3653), 50-55. https://doi.org/10.1126/science.147.3653.50

Redfield, A. (1972). Development of a New England salt marsh. Ecological Monographs, 42(2), 201–237. https://doi.org/10.2307/1942263
Reed, D. J. (1995). The response of coastal marshes to sea-level rise: Survival or submergence? Earth Surface Processes and Landforms, 20(1), 39–48. https://doi.org/10.1002/esp.3290200105

Reimer, P. J., Bard, E., Bayliss, A., Beck, J. W., Blackwell, P. G., Ramsey, C. B., et al. (2013). IntCal13 and Marine13 radiocarbon age calibration curves 0–50,000 years cal BP. Radiocarbon, 55(4), 1869–1887. https://doi.org/10.2458/azu\_js\_rc.55.16947

Roy, P. S. (1994). Holocene estuary evolution—Stratigraphic studies from southeastern Australia. Incised-Valley Systems, 51, 241–263. https://doi.org/10.2110/pec.94.12.0241

Saintilan, N., Khan, N. S., Ashe, E., Kelleway, J. J., Rogers, K., Woodroffe, C. D., & Horton, B. P. (2020). Thresholds of mangrove survival under rapid sea level rise. *Science*, 368(6495), 1118–1121. https://doi.org/10.1126/science.aba2656

Sella, G. F., Stein, S., Dixon, T. H., Craymer, M., James, T. S., Mazzotti, S., & Dokka, R. K. (2007). Observation of glacial isostatic adjustment in "stable" North America with GPS. *Geophysical Research Letters*, 34, L02306. https://doi.org/10.1029/2006GL027081

Syvitski, J. P. M., Vörösmarty, C. J., Kettner, A. J., & Green, P. (2005). Impact of humans on the flux of terrestrial sediment to the global coastal ocean. *Science*, 308(5720), 376–380. https://doi.org/10.1126/science.1109454

Tarasov, L., & Peltier, W. R. (2002). Greenland glacial history and local geodynamic consequences. *Geophysical Journal International*, 150(1), 198–229. https://doi.org/10.1046/j.1365-246X.2002.01702.x

Törnqvist, T. E., Jankowski, K. L., Li, Y. X., & González, J. L. (2020). Tipping points of Mississippi Delta marshes due to accelerated sea-level rise. *Science Advances*, 6(21), eaaz5512. https://doi.org/10.1126/sciadv.aaz5512

Trimble, S. W. (1974). Man-induced soil erosion on the southern Piedmont, 1700-1970. Soil Conservation Society of America.

Trimble, S. W. (1977). The fallacy of stream equilibrium in contemporary denudation studies. *American Journal of Science*, 277, 876–887. https://doi.org/10.2475/ajs.277.7.876

Trimble, S. W. (1999). Decreased rates of alluvial sediment storage in the Coon Creek Basin, Wisconsin, 1975-93. *Science*, 285(5431), 1244–1246. https://doi.org/10.1126/science.285.5431.1244

Tweel, A. W., & Turner, R. E. (2012). Watershed land use and river engineering drive wetland formation and loss in the Mississippi River birdfoot delta. *Limnology and Oceanography*, 57(1), 18–28. https://doi.org/10.4319/lo.2012.57.1.0018

U.S. Fish and Wildlife (2014). National wetlands inventory. Retrieved April 3, 2014, from http://www.fws.gov/wetlands/

Walling, D. E. (2006). Human impact on land-ocean sediment transfer by the world's rivers. *Geomorphology*, 79(3-4), 192-216. https://doi. org/10.1016/j.geomorph.2006.06.019

Walter, R. C., & Merritts, D. J. (2008). Natural streams and the legacy of water-powered mills. Science, 319(5861), 299–304. https://doi.org/ 10.1126/science.1151716

Weston, N. B. (2014). Declining sediments and rising seas: An unfortunate convergence for tidal wetlands. *Estuaries and Coasts*, 37(1), 1–23. https://doi.org/10.1007/s12237-013-9654-8

Yousefi Lalimi, F., Marani, M., Heffernan, J. B., D'Alpaos, A., & Murray, A. (2020). Watershed and ocean controls of salt marsh extent and resilience. Earth Surface Processes and Landforms, 45(6), 1456–1468. https://doi.org/10.1002/esp.4817

#### **References From the Supporting Information**

American Society of Testing and Materials (1991). Annual book of ASTM standards. Philadelphia, Pennsylvania: American Society for Testing and Materials.

Conservation Biology Institute (2010). Global average annual surface runoff, 1950–2000. Retrieved January 1, 2017, from http://www.arcgis.com/home/item.html?id=c5a76c73d0874e4c8d5bdb3bd40f2ed8

Kemp, A. C., & Haven, N. (2013). Radiocarbon dating of plant macrofossils from tidal-marsh sediment. Treatise on geomorphology (Vol. 14, pp. 370–388). San Diego, CA: Academic Press. https://doi.org/10.1016/B978-0-12-374739-6.00400-0

McCormick, C. L. (1968). Holocene stratigraphy of the marshes at Plum Island. Massachusetts. Amherst: University of Masschusetts.

McIntire, W. G., & Morgan, J. P. (1962). Recent geomorphic history of Plum Island, Masschusetts and Adjacent Coasts. Baton Rouge: State University Press.

National Oceanic and Atmospheric Adminstration (2016). VDatum Software. Retrieved February 24, 2017. from https://vdatum.noaa.gov/ welcome.html

U.S. Environmental Protection Agency, & United States Geological Survey (2012). National Hydrography Dataset Plus Version 2. Retrieved June 30, 2014, from http://www.horizon-systems.com/nhdplus/